Author's response to reviews

Title: Efficacy of multiparametric telemonitoring on respiratory outcomes in elderly people with COPD: a randomized controlled trial.

Authors:

Claudio Pedone (c.pedone@unicampus.it)
Domenica Chiurco (d.chiurco@unicampus.it)
Simone Scarlata (s.scarlata@unicampus.it)
Raffaele Antonelli Incalzi (r.antonelli@unicampus.it)

Version: 4 Date: 4 July 2012

Author's response to reviews:

Dear Editor --

The required changes have been made in the Methods (added the Ethical Committee specification at the end of the "Study Design" paragraph) and in the Acknowledgements (conflict of interest statement).

Best regards

Claudio Pedone